Author's response to reviews

Title: Perceptions of consent, permission structures and approaches to the community: a rapid ethical assessment performed in North West Cameroon.

Authors:

Jonas A Kengne-Ouafo (arnaudkengne@yahoo.com)
Theobald M Nji (muenji@yahoo.com)
William F Tantoh (farnyutantoh@yahoo.com)
Doris N Nyoh (dnnamnyoh@yahoo.com)
N Tendongfor (ntendongfor@yahoo.com)
Peter A Enyong (enyongpa@gmail.com)
Melanie J Newport (m.j.newport@bsms.ac.uk)
Gail Davey (g.davey@bsms.ac.uk)
Samuel Wanji (swanji@yahoo.fr)

Version: 4 Date: 21 September 2014

Author's response to reviews: see over
Dear editor,

We have once more the pleasure to re-submit the revised version of our manuscript entitled "Perceptions of consent, permission structures and approaches to the community: a rapid ethical assessment performed in North West Cameroon". All the concerns raised have been addressed and the corrections effected in the text. The map depicted in figure 1 was developed by our research team. The paper has been formatted according to the journal style and adheres to the RATS guidelines for reporting qualitative studies.

Yours sincerely,

Samuel Wanji, PhD